Author’s response to reviews

Title: The impact of dexmedetomidine added to ropivicaine for transversus abdominis plane block on stress response in laparoscopic surgery: a randomized controlled trial

Authors:
Zhaojun Qin (qin1172001@aliyun.com)
Chunyan Xiang (1325186616@qq.com)
Hongbo Li (707792268@qq.com)
Tingting Liu (835568506@qq.com)
Leyun Zhan (Leyunzhan@163.com)
Zhengyuan Xia (zyxia@hku.hk)
Min Zhang (30506222@qq.com)
Jianpin Lai (279522080@qq.com)

Version: 1 Date: 15 Feb 2019

Author’s response to reviews:

1 It is noticed that there are some textual overlaps with previously published papers, in particular:
   line 92-94; line 97-99
Please rephrase those sentences to reduce textual overlap.
Reply: We have rephrased these sentences in our revised manuscript.

2 At this stage, we ask that you submit a clean version of your manuscript and do not include track changes or highlighting.
Reply: We have deleted the highlighting in our revised manuscript.